Author's response to reviews

Title: Knowledge about AIDS and HIV Risk-related Sexual Behavior among Nigerian Naval Personnel

Authors:

Ugboga A Nwokoji (unwoo@yahoo.com)
Dr Ademola J Ajuwon (ajajuwon@yahoo.com)

Version: 5 Date: 18 May 2004

Dear Editor,

Please find attached, a revision of our manuscript, titled, Knowledge of AIDS and risk-related sexual behavior among Nigerian naval personnel. We have made the following revisions on the paper:

1. All unnecessary capitalisation have been removed from the title,
2. The full contact details of each of the two authors have been provided using a superscript formal (1, 2)
3. The tables have been reconstructed with rows and borders to conform with the editorial policy of BMC. All unnecessary capitals letter have been removed from the tebales
4. All typographical errors detected have been corrected.

Yours sincerely,
Dr. Ademola J. Ajuwon